Author's response to reviews

Title: Antibacterial resistance and their genetic location in MRSA isolated in Kuwait hospitals, 1994-2004

Authors:

Edet E. Udo (EDET@has.edu.kw)
Noura Al-Sweih (nalsweih@hsc.edu.kw)
Eiman Mokaddas (e.mokaddas@hsc.edu.kw)
Molly Johny (johnymolly@hotmail.com)
Rita Dhar (rtadhar50@hotmail.com)
Huda H. Gomaa (hodahelmi@hotmail.com)
Inaam Al-Obaid (inaob218@hotmail.com)
Vincent O Rotimi (vincent@hsc.edu.kw)

Version: 4 Date: 23 November 2006

Author's response to reviews: see over
23 November, 2006

Editor in-Chief
BMC Infectious Disease
BMC –series journal
Biomed Central

Re: formatting changes for Manuscript: BMC MS # 1505310463114520

Antibacterial resistance and their genetic location in MRSA isolated in Kuwait hospitals, 1994-2004 by Edet. E. Udo, Noura Al-Sweih, Eiman Mokaddas, Molly Johny, Rita Dhar, Huda H Gomaa, Inaam Al-Obaid and Vincent O. Rotimi.1

Dear Sir,

I have made the suggested changes to the manuscript as follows.

1. The full stop at the end of the title has been removed.
2. The keywords section has been removed
3. The Material and methods has been changed to Methods
4. The figures have been cropped and labeled figures 1 and 2 since a composite picture could not be uploaded to the system. Corresponding changes have been made in the ext.

All authors have approved the revised copy of the paper.

Yours sincerely

Edet E. Udo PhD
Department of Microbiology
Faculty of Medicine, Kuwait University
P. O. Box 24923, Safat, 13110
KUWAIT

Fax: + (965) 533 2719

Email: EDET@hsc.edu.kw